Author's response to reviews

Title: Injuries in Syria; first population-based estimates and characterization of predominant types

Authors:

Wasim Maziak (maziak@net.sy)
Kenneth D Ward (kdward@memphis.edu)
Samer Rastam (samer@scts-sy.org)

Version: 2 Date: 13 June 2005

Author's response to reviews:

Title changed, and abstract revised